Author's response to reviews

Title: Minimally Invasive Scoliosis Surgery: An Innovative Technique in Patients with Adolescent Idiopathic Scoliosis

Authors:

Vishal Sarwahi MD (vsarwahi@montefiore.org)
Adam L Wollowick MD (adwollow@montefiore.org)
Etan Sugarman MD (etan.sugarman@med.einstein.yu.edu)
Jonathan J Horn BS (johorn@montefiore.org)
Melanie Gambassi cPNP (MGAMBASS@montefiore.org)
Terry D Amaral MD (tamaral@montefiore.org)

Version: 4 Date: 5 August 2011

Author's response to reviews: see over
The following revisions have been carried out:

Major revisions (we require the author to make these changes)

Competing interests: Please include a 'Competing interests' section between the Conclusions and Authors contributions. If there are none to declare, please write 'The authors declare that they have no competing interests'.

- Section on competing interests have been added.

Authors’ contributions: Please place the Authors’ contributions section towards the end of the manuscript text, directly preceding the Acknowledgements.

- Section on authors’ contributions have been added.

Figures: Please provide figure titles. These should be concise and be the first sentence of the figure legend.

- Figure titles have been added prior to each figure.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Author List: Please do not include any professional titles in the author list in the online submission system.

- None were included

Line Numbers: Please remove all line numbers present in the manuscript.

- Line numbers were removed

Sections Headings: Please do not include bullet points in your section headings.

- Bullets have been removed

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

- We believe all typographical errors to have been removed. Thank you very much!